Author's response to reviews

Title: Seroepidemiology of Toxoplasma gondii infection in women of child-bearing age in Central Ethiopia

Authors:

Endrias Z. Gebremedhin (endrias.zewdu@gmail.com)
Anteneh H Abebe (antenhailu@yahoo.com)
Tesfaye S Tessema (tesfu74@yahoo.com)
Kassu Desta Tullu (kassudesta2020@gmail.com)
Girmay Medhin (gtmedhin@yahoo.com)
Maria Vitale (marvitus@yahoo.com)
Vincenzo Di Marco (dimarcov@email.it)
Eric Cox (eric.cox@ugent.be)
Pierre Dorny (pdorny@itg.be)

Version: 2 Date: 24 April 2012

Author's response to reviews: see over
Dear Sir/Madam:

I have sent the necessary documents as per your requirement for consideration of publication in BMC Infectious Disease journal. Our article entitled “Seroepidemiology of Toxoplasmosis in women of child bearing age in Central Ethiopia” contains useful information about seroprevalence and risk factors to acquire *T. gondii* infection by women of child-bearing age which are important in combating the disease, hence worth publication. I believe that the contents of the article fit well into the topics covered by the journal. I also would like to declare that all authors have no competing interests. Finally, on behalf of the authors of the article, I would like to mention that we are glad to send our work to BMC Infectious disease journal which is open access, highly valued and cited journal.

I am very much grateful for entertaining my inquiries.

With best regards

Endrias Zewdu (PhD cand.)
Addis Ababa University
Ethiopia